# BANCA D'ITALIA

# Temi di discussione

del Servizio Studi

# Do capital gains affect consumption? Estimates of wealth effects from italian households' behavior

by Luigi Guiso, Monica Paiella and Ignazio Visco



Number 555 - June 2005

The purpose of the Temi di discussione series is to promote the circulation of working papers prepared within the Bank of Italy or presented in Bank seminars by outside economists with the aim of stimulating comments and suggestions.

The views expressed in the articles are those of the authors and do not involve the responsibility of the Bank.

*Editorial Board:* Giorgio Gobbi, Marcello Bofondi, Michele Caivano, Andrea Lamorgese, Francesco Paternò, Marcello Pericoli, Alessandro Secchi, Fabrizio Venditti, Stefania Zotteri. *Editorial Assistants:* Roberto Marano, Cristiana Rampazzi.

# DO CAPITAL GAINS AFFECT CONSUMPTION? ESTIMATES OF WEALTH EFFECTS FROM ITALIAN HOUSEHOLDS' BEHAVIOR

by Luigi Guiso<sup>\*</sup>, Monica Paiella<sup>\*\*</sup> and Ignazio Visco<sup>\*\*</sup>

#### Abstract

We use detailed data on housing prices in Italy available for a large number of years and with a fine geographical breakdown to compute capital gains and losses on the most widespread asset among consumers, housing, and inquire whether changes in housing values affect consumption. We find that consumer expenditures do react to capital gains, with a marginal propensity to consume out of real value changes of housing wealth of about 0.02. Reactions are different across types of consumers: while homeowners increase consumption when house prices increase, with a marginal propensity of about 0.035, the renters' response to the higher house cost tends to be that of increased savings. For the owners of listed stocks the response to capital gains is difficult to estimate with statistical precision, even if, for the limited sample of owners of these assets, its negative sign may be indicative of prevailing substitution over income effects.

JEL classification: D12, E21, E44.

Keywords: wealth effects, consumption, housing, stock ownership.

#### Contents

1. Introduction	7
2. Literature Review	9
3. Data Sources and Description	12
3.1 Computing capital gains	
4. The Model	
5. Entire Sample Result	
6. Homeowners, Home-Renters and Owners of Risky Financial Assets	
6.1 The effect of house prices on homeowners' consumption	
6.2 The effect of house prices on renters' consumption	
6.3 The reaction of consumption to capital gains on risky financial assets	22
7. Discussion and Conclusions	
Tables and Figures	
Appendix: definitions of financially constrained firms	
References	39

<sup>\*</sup> Università degli Studi di Sassari and Ente L. Einaudi.

<sup>\*\*</sup> Bank of Italy.

# **1. Introduction**<sup>1</sup>

Whether and how much changes in consumers' wealth affect their consumption decisions is relevant for a number of reasons, including a better understanding of how asset prices impact on the economy and how monetary policy works. Following the seminal contributions of Modigliani, Brumberg and Ando<sup>2</sup> of about half a century ago, there has been a lively and important debate on the nature of the relationship between wealth and consumption and on the timing of the "wealth effects" that changes in asset prices may have on consumption. More recently, the question about the relevance of wealth effects in consumption has been raised in connection with the stock market boom of the 1990s and the slump of the last few years.<sup>3</sup>

In this paper we study the effects of changes in the market price of consumer assets using micro data on Italian households available biannually for more than a decade. We isolate two sources of wealth re-valuations, changes in the market value of housing and changes in the market value of stocks, but focus primarily on the former. Whether the dramatic changes in stock prices that have been observed during the 1990s have affected consumption or not is of primary importance to understand the business-cycle effects of stock market booms and crashes. Many studies have indeed focused on the effect on consumption of capital gains on equities. However, stocks are only one of the assets in which consumers invest their savings. Furthermore, aside from the United States where half of the population invests in stocks, in most countries equities are only held by a small fraction of wealthier households (Guiso, Haliassos and Jappelli, 2001), whose consumption is possibly less responsive to changes in the market value of wealth. On the contrary, real

<sup>&</sup>lt;sup>1</sup> This paper has been prepared for a volume in memory of Albert Ando. We thank Giuseppe Grande and Francesco Zollino for providing us with the asset price data. The views expressed in this paper are those of the authors and do not necessarily reflect those of the Bank of Italy. E-mail for correspondence: monica.paiella@bancaditalia.it.

<sup>&</sup>lt;sup>2</sup> See especially Modigliani and Brumberg (1954 and 1980), Modigliani and Ando (1960) and Ando and Modigliani (1963). For a discussion of the relation between the life cycle hypothesis advanced by Modigliani and his co-authors and the permanent income hypothesis advanced by Friedman (1957), see Deaton (1992).

<sup>&</sup>lt;sup>3</sup> See Poterba (2000) for a survey of the time-series evidence and Case, Quigley and Shiller (2001) for a cross-country comparison. Boone and Girouard (2002) and Bayoumi and Edison (2003) provide macroeconomic estimates of wealth effects based on panel data for several countries.

estate and housing investment is widespread and for many households it is a substantial portion of their wealth. In addition, housing prices vary considerably, imparting significant changes to housing net worth.

Focusing on real estate has also an additional advantage with respect to studies of stock capital gains. Most surveys on household portfolios only report whether households own stocks and what is their overall amount, but they do not usually provide detailed information on the type of stocks held. Therefore, in order to compute capital gains one has to assume that each household holds the market portfolio and use stock market indexes to compute price changes in individual portfolios. Thus, capital gains vary across households only with their overall level of stockholdings. This is bound to introduce measurement error as, on the contrary, stockholders generally invest in a small number of stocks (e.g. King and Leape, 1998) and individual stock prices vary considerably more than the stock market index. As a consequence, capital gains computed at the household level may not be a good proxy for actual capital gains, leading to attenuation bias in the estimated effect on consumption. These problems are not shared by our data on housing prices. We use, in fact, detailed information on housing prices, available for several years and with a fine geographical breakdown. This allows us to match house prices with the household-specific real estate and to compute capital gains and losses at the household level. Thus, capital gains differ across agents both because households differ in the amount of real estate that they own, so that consumers with larger housing wealth are more exposed to house price variation, and because house prices vary across locations and types of estate. The combined time-series and cross-sectional variation in capital gains potentially allows to pin down their effect on individual consumption with greater precision than that allowed by time variation alone in aggregate stock prices.

One important issue when assessing the effect of capital gains on consumption is to distinguish between their effect on consumption at the individual and at the aggregate level. Strong response of individual household consumption to capital gains may be consistent with weak or no response at the aggregate level if capital gains for a group of consumers represent capital losses for another. Take the case of housing: at each point in time not all consumers are homeowners. With limited access to the mortgage market some households are saving to purchase a house in the future. Others, most likely older households, have saved in the past and now own the asset. An increase in the relative price of housing causes

an increase in the wealth of homeowners that may translate in higher consumption. It may however result in lower consumption for current home-renters, but potential home-buyers, as they now need more money to become homeowners. Reliance on aggregate time series to detect an effect of capital gains on consumption clearly fails to assess heterogeneous responses by different groups of households to the same asset price movement and may lead to the wrong conclusion that consumption does not, or weakly, respond to capital gains. In particular, if aggregate consumption were found not to respond to capital gains, it would not be possible to assess whether this would be the result of consumers not changing their expenditures when faced with changes in the market value of their wealth holdings, or because of heterogeneous responses that cancel out in the aggregate. With micro data we can directly address this issue.

One of the main findings of this paper is that, considering the entire sample of households, they raise their consumption by slightly more than 2 cents for each euro of capital gains on housing. However, responses differ among owners and renters: while homeowners increase consumption with a marginal propensity to consume out of real value changes in housing wealth that is close to 0.035, in line with the evidence from other studies for the United States and the United Kingdom, renters, if anything, appear to decrease it, counteracting the effect on aggregate consumption, even if their response cannot be estimated with statistical precision.

The rest of the paper is organized as follows. Section 2 reviews the empirical literature on consumption and capital gains. Section 3 describes the data and the procedure used to compute the capital gains. Section 4 outlines the general model that guides us in our empirical analysis and Section 5 shows the main empirical results. Section 6 extends the estimates in various directions. Section 7 summarizes and concludes.

#### 2. Literature Review

Most recent empirical research on the link between wealth and consumption has found evidence of a positive and significant long-run relationship between the two variables. Indeed, budget constraints imply that an increase in household wealth must eventually raise its expenditure. The "direct" consumption response might be more or less rapid depending on the uncertainty about the permanence of the asset price change and more or less strong depending on the (real or perceived) accessibility to the asset in question, which could be held to provide for retirement, bequest, or other motives.

A positive relationship between wealth and consumption might also arise if changes in asset prices are considered helpful to predict changes in future income. As this effect may also account for the changes in the marginal propensity to consume out of wealth on the part of those who do not own the particular assets to which the price changes refer, it is often referred to, in the more recent literature, as the "indirect channel". More generally, an asset price effect on consumption was also early derived within the theoretical framework of the life cycle hypothesis. In the words of Franco Modigliani (1975, p. 15), long-term changes in the real rates of return of net assets would affect "the life profile of consumption both through an income effect, tending to raise consumption at all ages, and a substitution effect tending to shift consumption away from the present and towards a later age. [...] Unfortunately there is no way to establish a priori just how strong the substitution effect might be in reality". In principle, then, rather than guessing at how capital gains and losses might "indirectly" affect the propensity to consume, an attempt at directly estimating their effect on consumption would be possible if data on household asset price changes were available. One should expect that the higher the income effect with respect to the substitution effect, the larger would be the response of consumption to asset price changes (perceived to last).<sup>4</sup>

To disentangle all these effects household-level data are needed. Based on the US Survey of Consumer Finances, Poterba and Samwick (1995) highlight the case for an indirect channel (if any) by finding very limited correlation between stock prices and expenditure on luxury goods, which are consumed disproportionately by higher-income, stockholding households and whose share in total consumption should therefore increase in the wake of rising stock prices. Stockholders' spending appears to be only modestly affected

<sup>&</sup>lt;sup>4</sup> For an early extension in this direction of the time-series estimate of the Ando and Modigliani (1963) consumption function, see Arena (1964). A thorough discussion of the effect on consumption of capital gains and interest rate changes within the life cycle framework is that by Tobin and Dolde (1971), that builds on an earlier contribution by Tobin (1967). In particular, it should be observed that capital gains and losses may be due to changes in relative prices that do not involve interest rate movements, and may give rise to direct income effects on consumption, as well as to changes in the discount rates at which prospective asset earnings are capitalized, and which possibly induce substitution between present and future consumption. Under certain conditions one may argue that the former are more likely in the case of changes in real estate prices and the second may matter more in the case of stock prices.

by changes in wealth also on the basis of the qualitative evidence from the University of Michigan's *SRC Survey of Consumers* reviewed by Starr-McCluer's (2000). Empirical assessments of the Consumption-based Capital Asset Pricing Models (Mankiw and Zeldes, 1991, Attanasio, Banks and Tanner, 2002, Paiella, 2004, Vissing-Jørgensen, 2002, among others), based on household-level data for the United States and the United Kingdom, find instead that stockholders' expenditure is more highly correlated with stock market returns than non-stockholders', which supports the hypothesis of a direct effect. Coherent with this, Maki and Palumbo (2001) find that those households that benefited the most from the recent run-up in equity prices were also the group that substantially decreased their rates of saving. Further, Dynan and Maki (2001) find relatively large direct wealth effects, on the basis of household-specific measures of capital gains. Taken together, these results corroborate a direct view of the stock market wealth effect on consumption.

Even if the stock market wealth effect on consumption is direct, recent research on consumer behavior has suggested many reasons why consumers might increase their spending by less than simple calculations of the marginal propensity to consume over the life cycle may suggest. First of all, equity is held by relatively few, high-income households, which might reduce the aggregate response to this component of households' net worth<sup>5</sup>. Secondly, in the 1990s, there has been in several countries an increase in the importance of equity investment held in retirement accounts. If households develop "mental accounts",<sup>6</sup> that make them more likely to consume wealth held in some ways rather than in others, the marginal propensity to consume out of wealth gains accrued to retirement accounts is likely to be smaller than the marginal propensity to consume out of directly held assets, since the former are often thought of as "long-term assets". Finally, the amount by which capital gains affect spending may well be a function of whether or not the gain has been realized. In principle, unrealized gains can be borrowed against, through, for example, home mortgage refinancing, or could induce households to finance additional expenditure by selling other

<sup>&</sup>lt;sup>5</sup> Analytic results by Carroll and Kimball (1996) and numerical simulations by Zeldes (1989) show that, when uncertainty is introduced in the life cycle model, the consumption function becomes concave. Thus, the marginal propensity to consume out of wealth is lower for households with more resources. Empirical support for such concavity is found by Parker (1999), who estimates the relationship between consumption and liquid assets, and also by Dynan, Skinner and Zeldes (2000), who show that the *average* propensity to consume declines with permanent income.

<sup>&</sup>lt;sup>6</sup> See Tahler (1990) for details on this view.

assets or by reducing the marginal propensity to save out of current income. However, as long as the gain is not realized, it remains exposed to price uncertainty, which suggests that the propensity to spend out of unrealized gains is likely to be smaller than the propensity to spend out of realized gains. Related to this, it must be remembered that it is only the permanent shocks that matter for consumption decisions. In fact, as consumption is by definition irreversible, it would be unwise to consume this year a gain on an asset that next year may produce a loss.

Also the effect of housing wealth on consumption has been explored in some studies on the basis of data at the household level. Skinner (1996) finds a significant, although small, effect of housing wealth on consumption in the Panel Study on Income Dynamics (PSID), whereas Levin (1998) finds essentially no effects in the Retirement History Survey. Sheiner (1995) explores the implications of house price increases on PSID renters' savings, but her results are quite inconclusive. Engelhardt (1996) estimates the marginal propensity to consume out of capital gains in owner-occupied housing to be about 0.03 in the PSID, but this comes from an asymmetry in behavioral response, with those experiencing real capital losses reducing their expenditures and those experiencing gains leaving them unmodified. Always on the PSID, Hoynes and McFadden (1997) investigate the link between savings and capital gains in housing and find little evidence of a change in saving behavior in response to expectations about owner-occupied housing, consistent with the hypothesis of mental accounts of Thaler (1990). All these studies based on micro data rely upon owners' estimates of housing values. Although their bias may be small (see Kain and Quigley, 1972, and Goodman and Ittner, 1992), owners' estimates typically have high sampling variances, which leaves much ambiguity in the interpretation of the results. More recently, using individual household data from the British Household Panel Survey and changes in market values at county levels, Disney, Henley and Jevons (2003) estimate a marginal propensity to consume out of shocks to housing wealth of around 0.04 to 0.08. Though somewhat different in focus, this study is close in spirit to our own and obtains results broadly similar to ours.

#### 3. Data Sources and Description

Our main data set is obtained from the Survey of Households Income and Wealth (SHIW), a large-scale household survey run biannually by the Bank of Italy on a random

sample of about 8,000 Italian households.<sup>7</sup> The survey is available for several years and covers at least two full business cycles with large fluctuations in assets prices; we rely on the 1991, 1993, 1995, 1998, 2000 and 2002 waves whose questionnaire contents, survey methodology and variable definitions are broadly homogeneous. Pooling them together we obtain a sample of over 47,500 records. The survey has a rotating panel component, so that about half of the observations refer to households which have been interviewed in more than one survey. The rotating component could prove useful in eliminating unobserved heterogeneity when studying the dynamics of consumption in response to wealth changes, a feature, however, that we do not exploit in this study.

The SHIW collects detailed information on Italian household income, consumption and wealth, as well as on households' portfolio allocation across financial instruments and their access to formal and informal credit. It has also detailed information on real estate wealth, including year of asset acquisition, size (in square meters) of the house, whether it is the main residence, property status and location. It should be observed that especially on financial wealth the survey participants appear to significantly under-report their assets (see on this issue D'Alessio and Faiella, 2002a). For each household, the Survey also provides information on characteristics of the households' head and each household member, such as education, age, place of birth, and residence. The Appendix contains a more detailed description of the data set and of the variables that we use in our study. Overall, our sample consists of 45,173 observations on 28,825 households.

Table 1 reports summary statistics for our set of data. Panel A shows demographic characteristics in three representative years (1993, 1998 and 2002) for the whole sample, for the sub-sample of real estate holders and for the sub-sample of risky financial assets holders for which capital gains either on real or financial assets are relevant. Real estate includes lands and buildings; risky financial assets are defined as the sum of listed and non listed equity shares, corporate bonds, managed investment accounts, and mutual funds. Characteristics are fairly constant over time, with the exception of the average household size that shrinks somewhat as a consequence of Italy's demographic changes. Differences in

<sup>&</sup>lt;sup>7</sup> This survey has been widely used in studies on saving behavior by Italian households. See, among others, the essays in the volume edited by Ando, Guiso and Visco (1994), and, for a description and assessment of the survey, Brandolini and Cannari (1994).

characteristics across groups are also stable: while the average real estate holder looks similar to the average sample consumer, the average holder of risky financial assets is definitely more educated, somewhat younger and less likely to live in the South of Italy. Panel B shows median values of consumption expenditure on non-durable goods and services, earnings and asset holdings for the whole sample as well as for the groups of real estate holders and of risky asset holders. As expected, assets holders, particularly those holding risky financial assets, are relatively better off in terms of both earnings and consumption, and wealth. Depending on the year, risky assets holders are between two and three times wealthier than the median sample consumer.

In order to compute capital gains on real estate holdings we merge the household data with data on house prices taken from a database collected since 1965 by *Il Consulente Immobiliare*, a publication edited by *Il Sole 24ore*.<sup>8</sup> The data are collected twice a year at the province level<sup>9</sup> and report the average price per square meter of new or recently renovated dwellings located in three different areas: central city, semi-central area and the outskirts. Clearly, levels differ depending on the city location, although the dynamics of prices within a given province – which is what is relevant for computing capital gains – is fairly similar. For our analysis we merge the data using the code of the province where the household lives. Besides house prices, in order to compute capital gains on equity we merge the household data also with data on an aggregate price index of the shares listed on the Italian Stock Exchange.<sup>10</sup>

Table 2 shows summary statistics for the dynamics of asset prices in the survey years, separately for financial assets and housing. 1993 and 1998 are years of strong capital gains on stocks, while year 2002 sees substantial capital losses. House price dynamics are reported separately for three geographical areas; clearly, house prices tend to co-move, driven by the business cycles. Yet, there are differences in the size of the price changes across areas, reflecting local factors. Furthermore, it turns out that house prices are not necessarily

<sup>&</sup>lt;sup>8</sup> *Il Sole 24ore* is the most widely read Italian financial newspaper. *Il Consulente Immobiliare* is targeted to real estate agents and intends to provide a benchmark for real estate evaluation.

<sup>&</sup>lt;sup>9</sup> Italy's territory is divided into 20 regions and each region is divided into provinces whose size is about that of a US county. The number of provinces has been evolving over time; they were 95 at the beginning of the 1990s and have increased to 103 at the end of the past decade.

<sup>&</sup>lt;sup>10</sup> The data source is Bank of Italy up to 1993 and Morgan Stanley (MSCI Italy) after then.

synchronized with stock prices. For instance, in 1998, while the stock market was booming, house prices were flat. House price changes are averages of changes across provinces. As the high standard errors reported in brackets show, there is considerable dispersion in house price dynamics across local markets. Figure 1 shows the distribution across provinces of house price changes in the years covered in our data set: it is clear that there is considerable geographical variation both in sign and size. Thus, while stock prices vary only at the aggregate level and we cannot account for differences in the valuation of individual portfolios as no information is available on the specific stocks held by households, the data on house prices vary over time, over space (across the Italian provinces) and across the location of the house (central versus semi-central versus outskirts). Accordingly, for the latter there is considerable variation and thus ample possibilities of identifying wealth effects, if present.

#### 3.1 Computing capital gains

Survey information on assets held and their value is reported as of the end of the year of interview. To obtain an estimate of the total net wealth as of the beginning of the year, we subtract household savings – computed as difference between disposable income and consumption expenditures – from household end-of-period reported net worth. To recover beginning-of-period net real and financial wealth we assume that household savings are distributed between financial and real assets as in the national accounts of the corresponding year. All these beginning-of-period wealth measures are inclusive of capital gains/losses. To disentangle the latter, we assume that they accrue only on real estate (lands and buildings) and on listed stocks. We also assume that the real estate share of real assets and the stock share of financial assets at the beginning of the year are the same as those at the end of the year. Using the year percentage change in the price of the relevant asset, net of the change in consumer prices, we can then recover the real gain/loss on the asset of interest.

Table 3 reports summary statistics on capital gains by type of asset and year. The table also reports the fraction of households that invest in the asset in our sample. Clearly, capital gains are measured with reference to the holders of the specific asset. While about 70 percent of households hold real estate (67 percent in 1991 and 73 percent in 2002), only a minority has risky financial assets, though participation has been markedly increasing over the 1990s with the spreading of the "equity culture". In fact the fraction of households with risky assets

has risen from 6 percent in 1991 to about 21 ten years later and that of stockholders from to 2 to 8 percent (last rows). Average capital gains/losses reflect both the change in the price of the asset and the initial amount of the asset held among holders. In the aggregate, also the extensive margins - that is the fraction of consumers that have the asset in their portfolio matter: clearly, assets that are held by a minority of consumers are unlikely to have large effects on aggregate consumption. In this respect, the table makes clear that focusing on real estate rather than on stocks may be more important, at least when considering the Italian experience. The median value of capital gains on housing in 1991 is 4,200 euros and 8,900 euros in 2002 (all at 2002 prices), about 17 percent and 34 percent of median household earnings, respectively. In 1993 and 1995 the median household incurred capital losses on real estate, amounting to 2,700 and 4,300 euros, respectively 12 percent and 18 percent of its earnings. As to financial assets, we compute both the gains on the stock of risky assets, assuming that the prices of all risky securities vary proportionally to our stock market index, and the gains on listed stocks, which are a subset of the risky asset holdings. The capital gains on risky financial assets have peaks in 1993 and 1998 with median value among asset holders of 9,200 and 9,800 euros, respectively 25 percent and 29 percent of those years' household earnings; in 2002 the median capital loss of 5,400 euros amounts to about 14 percent of earnings. The gains on listed stocks range from 7 percent of earnings in 1998 to -6 percent in 2002. The actual gains/losses can be expected to fall somewhere in between as Italian households hold relatively large amounts of mutual funds (mutual funds accounted for almost 40 percent of risky financial assets in 2002), whose composition is not reported and which here we classify as risky assets.

The size of these fluctuations in asset values is remarkable and may potentially translate into large changes in consumption if consumers react to them. Although, as noted earlier, house prices co-move, there are significant geographical differences in house price dynamics in response to local shocks. Thus, as illustrated in Figure 1 there are locations where house prices decline even in years when on average they increase. As a consequence, in the same year some households may enjoy capital gains and other incur capital losses on their real estate: this is a further reason why aggregate data may conceal significant consumption responses to capital gains. Table 4 shows the fraction of "losers" (those facing a decline in the value of the asset) and "gainers" (those facing an increase) for each sample year. Interestingly, even in years when the average house prices increase significantly, as in 2002, there is a non negligible fraction of households who lose (9 percent).

# 4. The Model

We estimate a simple consumption function based on a general version of a life cyclepermanent income model, where rational, utility-maximizing agents optimally allocate their resources to consumption over their entire life.<sup>11</sup> In this instance, consumer decisions are conditional on human wealth (which may be assumed to be roughly proportional to labor and other non-asset income) and on non-human (real and financial) wealth and the equilibrium behavior of consumers can be described using the following:

(1) 
$$\frac{c_{h,t}}{y_{h,t}} = b_0 + b_1 \frac{p_{h,t}^w w_{h,t}}{y_{h,t}} + b_2 \frac{cg_{h,t+1}}{y_{h,t}} + \boldsymbol{e}_{h,t},$$

where  $c_{h,t}$ ,  $y_{h,t}$ , and  $p_{h,t}^w w_{h,t}$  are household *h* non-durable consumption and non-asset income over the *t*<sup>th</sup> year and beginning-of-year wealth (valued at beginning-of-year prices).  $cg_{h,t+1}$  denotes the real capital gain/loss accrued over the year on beginning-of-period wealth

and is defined as:  $cg_{h,t+1} = p_{h,t}^{w} w_{h,t} \left( \frac{\boldsymbol{p}_{h,t+1}^{w} - \boldsymbol{p}_{t+1}^{c}}{1 + \boldsymbol{p}_{t+1}^{c}} \right)$ , with  $\boldsymbol{p}_{h,t+1}^{w}$  and  $\boldsymbol{p}_{t+1}^{c}$  denoting the changes in

asset prices and in consumer prices, respectively.  $e_{h,t}$  captures unobservable heterogeneity in consumption, relative to income.

In practice, in the estimation of the marginal propensities to consume out of wealth and out of capital gains/losses, we distinguish between real and financial assets and consider only the gains/losses accrued on real estate (lands and buildings) and on stock holdings. We also take into account household heterogeneity by including a set of household sociodemographic variables that allow for differences in the marginal propensity to consume out of income. In principle, any heterogeneity in the marginal propensities to consume out of wealth and out of capital gains, for which we do not control, will also end up being captured by these variables. However, including a full set of controls and allowing explicitly for heterogeneity in the marginal propensities to consume out of wealth and capital gains does

<sup>&</sup>lt;sup>11</sup> While there are well known differences, though much less than similarities, between the life cycle and the permanent income theories of consumption behavior, they are not relevant at the level of generality considered in this study. For a thoughtful discussion of empirical evidence on the life cycle hypothesis, see Ando (1988).

not change our results in any way.<sup>12</sup>

#### 5. Entire Sample Result

Table 5 shows the results of our basic OLS estimate. Standard errors have been corrected to account for heteroskedasticity using White's (1980) estimator of the covariance matrix. The left-hand side is the ratio of consumption to non-asset earnings (mostly labor income). Therefore, also the right-hand side variables measuring household net assets or capital gains/losses are all scaled by household earnings. The first column reports a simple regression where the consumption-income ratio is regressed on the net worth-income ratio only. Wealth is measured as of beginning of period and therefore does not include capital gains/losses accrued to real estate and stocks. This regression is meant to give an idea of the size of the marginal propensities to consume out of wealth and out of labor income, the latter being measured by the regression's constant. The marginal propensity to consume out of household wealth is 0.02 and is highly statistically significant. The marginal propensity to consume out of labor income is about 0.71 and is also estimated with high precision. Both marginal propensities are consistent with the predictions of a standard Ando-Modigliani life cycle model, on which this specification is based, even if the former is on the low side and the latter on the high side of the spectrum of the estimates generally found in the literature. In particular, the latter result is indicative of the possible presence of non-negligible liquidity constraints for a relevant fraction of the households included in our sample.

The second column adds to the specification a number of demographic characteristics, including age, gender, job status and type, and location – all referred to the household's head. These controls are meant to capture life cycle effects on the propensity to consume, as well as preference heterogeneity across consumers which may be correlated with demographics. We also control for family size, to account for scale economies in consumption. To capture business cycle effects we include a set of year dummies. The way control variables are introduced implies that they affect primarily the marginal propensity to consume out of labor income, since they shift the regression's constant. As a consequence, the marginal propensity to consume out of income is now household specific, with a mean of

<sup>&</sup>lt;sup>12</sup> The regressions are available upon request.

0.70 and a standard deviation of 0.14. The estimated marginal propensity to consume out of wealth is 0.021. Both estimates are very close to those in the first column.

The third column adds to the regressions the capital gains and losses on real estate while controlling for beginning-of-period wealth. The marginal propensity to consume out of a capital gain on real estate is 0.019, only slightly smaller than the marginal propensity to consume out of total beginning-of-period wealth. The parameter is precisely estimated and we reject the hypothesis that is equal to zero. Yet, these estimates are run on the total sample which includes both holders and non-holders of real estate. If wealth effects are direct, they apply only to the former and including the latter may bias the estimates against finding an effect. Thus, in the fourth column we re-run the regression on the sample of households who hold real estate (land or buildings), which leaves us with 32,325 observations. Capital gains have now a larger positive and significant effect on consumption: one euro of increase in real estate value raises consumption by 2.8 cents. This estimate is similar to the one obtained by Engelhardt (1996) for the United States, though in the latter case it came out from an asymmetry in behavioral response, with those experiencing losses reducing their expenditure and those experiencing gains leaving it unmodified. It is also similar to, though a bit lower than, the estimate obtained by Disney, Henley and Jevons (2003) for the United Kingdom, who find a stronger response in periods of house price increase and a substantially higher marginal propensity to consume from real housing gains on the part of households with initial negative equity.

Since in the presence of liquidity constraints the marginal propensity to consume out of different forms of wealth may differ, the last two columns decompose wealth into real and financial. Indeed, we find that the marginal propensity to consume out of financial wealth is about twice that out of real assets (0.038 compared to 0.016, columns 5 and 6).<sup>13</sup> Disentangling the capital gains on real estate results in an estimated capital gain effect of 0.016, which is statistically significant. In sum, these estimates suggest that consumers do adjust consumption to perceived capital gains and losses on real estate. In the aggregate the response is by far smaller than that to one additional euro of labor and other non asset incomes, as it should be. The latter (a flow) is in fact included in the regression as a proxy of

<sup>&</sup>lt;sup>13</sup> As we noticed, however, financial wealth, and in particular the holdings of risky financial assets, is underestimated in the SHIW and this might induce an upper bias in the estimate of the relevant propensity.

the expected life-time earnings (a stock), a multiple, for the average household, of current earnings. This explains, in good substance, the difference between the propensities out of labor income and out of net housing and financial assets. In the case of capital gains, instead, we may conclude with Arena (1964, p. 109), ignoring possible substitution effects unlikely to affect the response to housing price changes, that "unanticipated gains have a 'one-shot' effect to be spread over the remaining life of the individual. They become part of net worth in future periods and therefore affect consumption in the same way as initial net worth …". Indeed, the marginal propensities to consume out of net assets and value changes of real estate are in our estimates of a comparable order of magnitude. Overall, the coefficients are somewhat on the low side, which may in part reflect the fact that our wealth variable is dominated by housing and the latter should only partly be considered, for those who live in the homes they own, as life cycle wealth. Also, a low wealth coefficient may be due to somewhat strong bequest motives that seem to play a significant role in Italy in the accumulation of wealth.<sup>14</sup>

#### 6. Homeowners, Home-Renters and Owners of Risky Financial Assets

So far in computing capital gains on real estate we have assumed that prices of second houses and land vary as the average house price of the province where the household lives. While it is likely that land prices are correlated with house prices, the correspondence is not necessarily one to one.<sup>15</sup> Strictly speaking our data apply well to homeowners. In this section we discuss results related respectively to the consumption decisions of homeowners and home-renters. We also briefly consider some evidence on the impact of changes in stock prices for the limited fraction of households that own risky financial assets.

# 6.1 The effect of house prices on homeowners' consumption

In Table 6 we report estimates based on the sample of 29,435 homeowners. They are obviously a sub-sample of real estate owners, as the latter may include renters that own for

<sup>&</sup>lt;sup>14</sup> See, for instance, Barca, Cannari and Guiso (1994).

<sup>&</sup>lt;sup>15</sup> Furthermore, the analysis relies on the assumption that the prices of the land and buildings other than the household's home vary as the prices of the dwellings of the province where the household resides, which needs

instance a piece of land. The household's home is priced based on the province and on its location within the province (central or semi-central location or outskirts). For the other real estate, we keep using the average provincial price. When estimates are conducted on this sub-sample, we find that the marginal propensity to consume out of wealth is 0.019 (first and second column), slightly smaller than the one that we obtain when estimates are done on the sample of real estate holders, while the estimated effect of capital gains on expenditure is 0.033. The marginal propensity to consume out of income is in the aggregate about 0.62, lower than for the entire sample. This would suggest that liquidity constraints are smaller for those who own the house where they live, which would seem to be a reasonable conclusion.

In the last column of Table 6 we distinguish between real and financial wealth. Homeowners' marginal propensity to consume out of financial wealth turns out to be substantially smaller than the marginal propensity for the sample as a whole, reported in the last column of Table 5. Furthermore, and in contrast with the evidence of Table 5, it is slightly smaller than their marginal propensity to consume out of real assets, although we cannot reject the hypothesis that the coefficients are equal. It would thus seem that those who only own their savings in the form of (possibly non risky) financial assets (such as bank and postal deposits or government bonds), being more liquidity constrained, should have higher propensities to consume out of these assets and of labor income. That this is the case is shown in the estimates of the consumption decisions of those who do not own real estate, which we now discuss.

# 6.2 The effect of house prices on renters' consumption

As said, changes in house prices may have heterogeneous effects on different groups of consumers. While owners may feel richer when house prices increase, renters who are saving to buy a house will feel poorer, as the purchasing power of their accumulated savings in units of houses declines with the price of housing. After the price increase, if renters do not give up their plans to become owners, they will need to save more and thus will compress current consumption. As a consequence, if present, this effect will weaken the effect of capital gains on aggregate expenditure caused by any positive effect on the

not be the case. However, there is no information as to the location of land and other buildings, with the exception of the 2002 survey that reports for the latter the region of location.

consumption of homeowners. We test this possibility in Table 7 where we report various estimates of the consumption equation for the group of renters. The first two columns focus on the sample of renters that have no real estate, which is the vast majority of renters (9,563 out of a total of 10,894). In the regressions we include the house price change in the province where the consumer lives, while controlling for year effects. Consequently, the identification of the effects of house price changes on renters' behavior stems from province-specific changes in house prices. As the estimates in the second column show, housing price increases have a negative effect on renters' consumption, though this coefficient estimate lacks statistical precision. Notice that, in accordance with our previous conclusion with respect to liquidity constraints, renters' marginal propensities to consume out of both wealth and income are substantially larger than those of homeowners and also of renters who own other real estate (last two columns of Table 7).

For the sample of renters who own other real estate the implications of house price changes are in principle less clear-cut. In fact, on the one hand, these households enjoy a gain on their real estate which may boost their consumption, on the other they still may suffer a loss if they are planning to buy a home. Thus, the overall effect of house price changes on this group is ambiguous. Columns 3 and 4 report consumption regressions for renters with other real estate, the latter including as an explanatory variable the estimate of the capital gain on their real estate. Though negative, also the estimate of the effect of house price dynamics on consumption for this group come with a rather large standard error.

#### 6.3 The reaction of consumption to capital gains on risky financial assets

To complete the evidence on the reaction of consumption to capital gains we have also computed capital gains on financial assets. As mentioned before, we have no information on the specific assets held in households' portfolios. To estimate household level capital gains on stocks, we assume that all listed equity moves as the stock market index, which is probably a source of measurement error in the estimated gains and losses, something that needs to be kept in mind when interpreting the results. Table 8 shows the results. The first two columns report regressions on the group of households that have neither stocks, nor other risky financial assets, such as bonds or mutual funds whose returns might be at least to some extent correlated to the return on stocks. If stock market effects are mainly direct, for this group stock market movements should not matter. Yet, as we see from the second column estimates, changes in the price index have a positive effect on their consumption. However, the coefficient may at least partly be capturing the correlation of the stock market with the business cycle, as this regression omits time dummies, which cannot be identified separately from stock price changes. Furthermore, the introduction of this variable also conflicts with the estimation of the response of consumption to capital gains and losses in real estate, especially so as independent cyclical effects are not identifiable.

The last three columns of the table focus on holders of risky financial assets. Overall capital gains on these assets seem to have a modest effect on consumption both from a statistical and an economic perspective. For the group of risky asset holders (third column), which includes also households who do not own stocks, the stock market index has a negative and significant sign, which may seem a puzzling result. However, equity price changes are most likely related to changes in interest rates, and this result might indicate that the substitution effect dominates in this case the income effect, a possibility suggested by Tobin (1967) and Tobin and Dolde (1971). When estimates are conducted on the (very small) group of stock holders (columns 4 and 5), the coefficient on the capital gains is negative but hardly significant. Besides the possibility that the substitution effect might substantially compensate the income effect, one further interpretation might be that consumers do not react to capital gains and losses on financial assets because these gains are highly volatile and consumption is irreversible: today's gains may be easily followed by losses tomorrow and a simple consumption smoothing logic would recommend little or no reaction. Unfortunately, the measurement problems discussed earlier do not allow us to separate this explanation from the competing one that the low estimates may also reflect attenuation bias due to measurement error. As to the marginal propensities to consume out of financial and real wealth and also out of income, the estimates for the risky asset holders turn out to be substantially smaller than those obtained for the sample as a whole. Indeed these consumers are the richest in the sample and the most likely to be very little affected by possible liquidity constraints.

# 7. Discussion and Conclusions

There are two main results from the estimates of consumers' reaction to capital gains on housing. The first is that Italian households do adjust consumption to changes in the valuation of their real estate with a response that is of similar size to that obtained in parallel studies for other countries: for the sample of those who own real estate, the marginal propensity to consume out of real value changes of housing wealth turns out to be about 0.035. The second result is that this reaction differs across consumers: when house prices change, what is a potential gain for real estate holders is a potential loss for those who still do not own a house but are saving to buy one. From the point of view of the economy as a whole, the relevant question is how important heterogeneous responses to house price changes are for aggregate consumption. Our estimates allow an answer to this question.

If house prices went up everywhere by 10 percent, for real estate holders the average gain would correspond to about 55 percent of their average annual real income. Based on the coefficients of column 4 in Table 5, their consumption (of non-durables and services) would increase by 2.3 percent (1.5 percent of earnings), that is, on average, 480 euros at 2002 prices (0.28 cents per euro invested in real estate). Considering only the homeowners (that is those households who own the house where they live), we find that their average increase in consumption is close to 3 percent (600 euros), based on the estimates of Table 6 (column 3). For the estimates obtained for the whole sample, however, the increase in consumption would be smaller, about 1 percent, according again to the results of Table 5 (columns 3 and 6). The latter obviously reflect the negative effect of an increase in house prices on renters. Indeed, for renters without any real estate, based on the estimates of column 2 in Table 7, the consumption-to-income ratio would drop by 6.3 percent, which would result in a 1,260 euro reduction in their average consumption, while for renters owning other real estate but not the house where they live the fall in consumption would amount (column 4 of Table 7) to about 2.7 percent (600 euros).

We should remind that in the case of renters the coefficient estimates of capital gains and losses on housing lack precision. However, aggregating these results for renters (23 percent of the sample) to that for homeowners (65 percent of the sample), and attributing no change in consumption to households who do not own nor rent the house where they live, we still get, for a 10 percent house price increase, an overall positive change of 105 euros, about 0.6 percent, in average consumption. On the other hand, given both the limited ownership by Italian households of risky financial assets as well as the insignificance of our empirical estimates, it can be reasonably expected that, somewhat differently than for other countries, the effect of equity price changes on overall consumption would be negligible.

In future work, it would be interesting to allow for different responses of consumption to permanent and transitory shocks to asset prices, on the presumption that only the former would matter for consumption decisions. In fact, as consumption is by definition irreversible, it would be unwise to consume this year a gain on an asset that next year may produce a loss. Our estimates reflect both the willingness of consumers to adjust consumption to variation in wealth as well as the time-series properties of the shocks to wealth valuation. As such, it is uncertain whether and by how much the response that we find may be limited by the presence of transitory shocks to house prices over the sample period. A filtering of transitory from permanent changes in real estate prices might allow to estimate with greater precision the response of consumers' to shocks perceived to be lasting. A further issue worth examining would be that of a possible asymmetric response of consumers to housing wealth losses and gains, as examined for example, with different results, by Engelhardt (1996) and Disney, Henley and Jevons (2003). Finally, in future work full advantage of the panel component of the SHIW, not exploited in this study, might be taken. This may lead to several improvements in the estimation of the effects of capital gains and losses on household expenditures. A crucial one would be that of making possible to model the dynamic response of consumers to shocks in permanent income and wealth. Taking better care of business cycle effects would also allow to estimate with greater precision the equilibrium response of consumption to wealth accumulation and capital gains and losses.

# **Tables and Figures**

# Table 1

# SUMMARY STATISTICS FOR THE SHIW SAMPLE

# A. Household characteristics

					Year				
		1993			1998			2002	
	All	Real estate	Risky asset	All	Real estate	Risky asset	All	Real estate	Risky asset
		Holders	Holders		Holders	Holders		Holders	holders
Age	53	53	50	53	54	51	55	56	53
Less than high	0.707	0.671	0.384	0.605	0.580	0.370	0.634	0.605	0.391
school*									
Household size*	3.025	3.114	3.137	2.978	3.032	2.976	2.716	2.783	2.859
Married*	0.741	0.782	0.843	0.743	0.777	0.801	0.685	0.723	0.782
Self-empl. head*	0.143	0.157	0.250	0.153	0.164	0.212	0.120	0.132	0.184
Not-working	0.449	0.444	0.292	0.447	0.452	0.357	0.532	0.550	0.422
Moved from region of birth*	0.202	0.182	0.193	0.180	0.162	0.178	0.191	0.169	0.197
Living in the South*	0.352	0.355	0.139	0.366	0.363	0.147	0.331	0.321	0.093
No. of obs.	7,730	5,440	753	6,793	4,988	1,374	7,484	5,478	1,556

Note: Real estate includes lands and buildings. Risky assets include stocks, private bonds and mutual funds. \* denotes "share of households". Mean values.

					Year				
		1993			1998			2002	
	All	Real estate	Risky asset	All	Real estate	Risky asset	All	Real estate	Risky asset
		Holders	Holders		Holders	Holders		holders	Holders
Earnings	22,200	26,100	41,200	23,800	26,800	36,300	23,700	27,200	36,700
Expenditure	16,400	18,400	25,500	16,900	18,000	23,800	16,800	18,000	24,000
Net wealth	98,200	147,300	294,000	105,400	145,700	222,800	114,800	162,800	241,300
Real assets	89,300	135,400	239,100	94,100	132,800	173,000	103,000	151,000	200,000
Real estate	-	134,400	-	-	121,900	-	-	150,000	-
Fin. assets	6,600	8,400	50,300	8,500	10,800	37,400	7,500	10,000	35,900
Risky fin. assets	-	-	33,000	-	-	36,500	-	-	20,000

*B.* Household income, consumption and wealth

Note: See Note to Panel A for the definitions of real estate and risky assets. Earnings and expenditure (on non-durables and services) refer to the year; the other variables are end-of-year values. Figures are in euros of year 2002. Median values.

				Year			
P(t+1)/P(t)		1991	1993	1995	1998	2000	2002
Stocks:		0.950	1.435	0.945	1.401	1.033	0.775
	North	1.066	1.031	0.996	1.010	1.035	1.084
		(0.110)	(0.070)	(0.095)	(0.048)	(0.058)	(0.064)
Houses:	Center	1.056	0.994	1.007	1.003	1.023	1.085
		(0.088)	(0.095)	(0.101)	(0.087)	(0.070)	(0.056)
	South	1.045	0.966	0.976	1.018	1.047	1.076
		(0.122)	(0.054)	(0.045)	(0.089)	(0.072)	(0.068)
	Italy	1.056	1.000	0.991	1.011	1.037	1.082
		(0.110)	(0.076)	(0.082)	(0.073)	(0.066)	(0.063)

## ASSET PRICES OVER TIME AND ACROSS REGIONS

Note: For stocks, the table reports the rate of growth of an index of the shares listed on the Italian Stock Exchange. For houses, the table reports the mean of the rate of growth of the provincial price per square meter and its standard deviation in parentheses. The ratios are at constant prices.

C	APITAL	GAINS O	N HOUSI	EHOLD A	SSETS			
		Year						
	1991	1993	1995	1998	2000	2002		
Real estate:								
Share of owners	0.673	0.704	0.716	0.734	0.738	0.732		
Capital gains	4,200	-2,700	-4,300	-1,000	1,600	8,900		
Capital gains/Earnings	0.170	-0.115	-0.180	-0.041	0.058	0.342		
Risky financial assets:								
Share of owners	0.064	0.097	0.113	0.202	0.231	0.208		
Capital gains	-900	9,200	-1,300	9,800	1,100	-5,400		
Capital gains/Earnings	-0.024	0.245	-0.037	0.285	0.032	-0.142		
Listed stock:								
Share of owners	0.018	0.033	0.031	0.077	0.101	0.082		
Capital gains	-300	2,600	-400	2,800	300	-2,800		
Capital gains/Earnings	-0.007	0.059	-0.012	0.069	0.008	-0.064		

# CADITAL CAINS ON HOUSEHOLD ASSETS

Note: The capital gains are computed by multiplying the stock of assets held at the beginning of the year by the change in the asset price over the coming year. See Note to Table 1, Panel A, for the definitions of real estate and risky assets. Figures are in euros of year 2002. Median values.

Table 4

THE LOSERS AND THE GAINERS Year 1991 1993 1995 1998 2000 2002 Real estate:

0.710

0.290

0.559

0.441

0.398

0.602

0.085

0.915

# Note: Losers and gainers are those real estate owners who recorded a capital loss or a capital gain on their assets, respectively.

0.684

0.316

0.261

0.738

Share of losers

Share of gainers

	(1)	(2)	(3)	(4)	(5)	(6)
	All	All	All	Real estate holders	All	All
(Net wealth)/Y	0.020*** (0.002)	0.021*** (0.003)	0.021*** (0.003)	0.021*** (0.002)		
(Net financial. wealth)/Y	(0.002)	(0.005)	(0.000)	(0.002)	0.038*** (0.010)	0.038*** (0.010)
(Net real wealth)/Y					0.016***	0.016***
Cap. gains on real estate/Y			0.019*** (0.007)	0.028*** (0.007)	(0.002)	(0.002) 0.016** (0.007)
Age/10		-0.118***	-0.119***	-0.024 (0.015)	-0.112***	-0.112***
Age squared/100		(0.023) 0.002 (0.002)	(0.022) 0.002 (0.002)	-0.002*	(0.022) 0.002 (0.002)	(0.022) 0.002 (0.002)
High school diploma		(0.002) -0.094***	(0.002) -0.095***	(0.001) -0.073***	(0.002) -0.092***	(0.002) -0.093***
University degree		(0.010) -0.210***	(0.010) -0.210***	(0.007) -0.157***	(0.010) -0.211***	(0.010) -0.212***
Household size		(0.012) -0.012***	(0.012) -0.012***	(0.009) -0.010***	(0.013) -0.010***	(0.013) -0.010***
Married		(0.003) -0.072***	(0.003) -0.071***	(0.002) -0.007	(0.003) -0.069***	(0.003) -0.069***
Self-employed head		(0.011) -0.011	(0.011) -0.011	(0.005) -0.032***	(0.010) 0.009	(0.010) 0.008
Not-working		(0.013) 0.193***	(0.013) 0.192***	(0.012) 0.094***	(0.011) 0.192***	(0.011) 0.192***
Moved from region of birth		(0.015) 0.066***	(0.015) 0.066***	(0.010) 0.030***	(0.015) 0.062***	(0.015) 0.063***
Living in the Center		(0.010) 0.036***	(0.010) 0.038***	(0.006) 0.037***	(0.009) 0.040***	(0.009) 0.042***
Living in the South		(0.008) 0.144***	(0.008) 0.145***	(0.005) 0.106***	(0.008) 0.147***	(0.008) 0.148***
Year: 1993		(0.009) 0.074***	(0.009) 0.081***	(0.006) 0.032***	(0.010) 0.078***	(0.010) 0.083***
Year: 1995		(0.011) 0.076***	(0.012) 0.084***	(0.008) 0.061***	(0.012) 0.081***	(0.012) 0.087***
Year: 1998		(0.007) 0.043***	(0.008) 0.048***	(0.007) 0.015*	(0.007) 0.044***	(0.007) 0.048***
Year: 2000		(0.009) 0.034***	(0.009) 0.037***	(0.008) 0.012*	(0.009) 0.034***	(0.009) 0.036***
Year: 2002		(0.008) 0.036***	(0.008) 0.033***	(0.007) 0.013**	(0.008) 0.035***	(0.008) 0.033***
Constant	0.705*** (0.011)	(0.007) 1.211*** (0.059)	(0.007) 1.208*** (0.059)	(0.006) 0.782*** (0.040)	(0.007) 1.191*** (0.057)	(0.007) 1.189*** (0.057)
Observations R-squared	45,173 0.057	45,173 0.093	45,173 0.094	32,325 0.215	45,173 0.102	45,173 0.103
Marginal propensity to consume out of income	0.705 (0.011)	0.698 (0.138)	0.697 (0.139)	0.603 (0.083)	0.713 (0.136)	0.712 (0.136)

# HOUSEHOLD MARGINAL PROPENSITY TO CONSUME OUT OF WEALTH

Note: The left-hand side variable is the ratio of household expenditure on non-durable goods and services to household earnings, denoted by Y. For the construction and definition of variables, see the data appendix. Robust standard errors in parentheses. \*significant at 10%; \*\* significant at 5%; \*\*\* significant at 1%.

	(1) All homeowners	(2) All homeowners	(3) All homeowners
(Net wealth)/Y	0.019***	0.019***	
(	(0.002)	(0.002)	
(Net financial wealth)/Y	()	()	0.015***
			(0.002)
(Net real wealth)/Y			0.020***
× ,			(0.002)
Cap. gains on real estate/Y		0.033***	0.034***
1.0		(0.007)	(0.007)
Age/10	-0.017	-0.017	-0.016
-	(0.012)	(0.012)	(0.012)
Age squared/100	-0.003**	-0.003**	-0.003***
C I	(0.001)	(0.001)	(0.001)
High school diploma	-0.071***	-0.072***	-0.072***
- 1	(0.007)	(0.007)	(0.007)
University degree	-0.152***	-0.153***	-0.152***
	(0.008)	(0.008)	(0.008)
Household size	-0.012***	-0.012***	-0.012***
	(0.002)	(0.002)	(0.002)
Married	-0.004	-0.003	-0.003
	(0.005)	(0.005)	(0.005)
Self-employed head	-0.027***	-0.028***	-0.032***
	(0.010)	(0.010)	(0.011)
Not-working	0.097***	0.096***	0.096***
_	(0.008)	(0.008)	(0.008)
Moved from region of birth	0.024***	0.024***	0.024***
	(0.005)	(0.005)	(0.005)
Living in the Center	0.031***	0.035***	0.034***
	(0.004)	(0.005)	(0.005)
Living in the South	0.102***	0.105***	0.104***
	(0.005)	(0.005)	(0.006)
Year: 1993	0.013**	0.030***	0.029***
	(0.006)	(0.007)	(0.007)
Year: 1995	0.039***	0.058***	0.058***
	(0.006)	(0.007)	(0.007)
Year: 1998	0.001	0.013*	0.013*
	(0.007)	(0.008)	(0.008)
Year: 2000	0.007	0.014**	0.015**
	(0.006)	(0.006)	(0.006)
Year: 2002	0.019***	0.014**	0.014**
	(0.006)	(0.006)	(0.006)
Constant	0.786***	0.773***	0.767***
	(0.032)	(0.032)	(0.032)
Observations	29,435	29,435	29,435
R-squared	0.208	0.212	0.213
Marginal propensity to	0.617	0.614	0.619
consume out of income	(0.080)	(0.082)	(0.081)

# HOMEOWNERS' MARGINAL PROPENSITY TO CONSUME OUT OF WEALTH

Note: See Note to Table 5 for the definition of variables. Robust standard errors in parentheses. \*significant at 10%; \*\* significant at 5%; \*\*\* significant at 1%.

	(1) No other real estate	(2) No other real estate	(3) Owners of other real estate	(4) Owners of other real estate
(Net wealth)/Y	0.062**	0.062**	0.035***	0.035***
$\mathbf{P}^{\mathrm{H}}_{t+1}$ / $\mathbf{P}^{\mathrm{H}}_{t}$	(0.027)	(0.027) -0.063 (0.067)	(0.009)	(0.009)
Cap. gains on real estate/Y				-0.084 (0.091)
Age/10	-0.049	-0.048	-0.188	-0.184
	(0.071)	(0.071)	(0.207)	(0.203)
Age squared/100	-0.008 (0.007)	-0.008 (0.007)	0.010 (0.018)	0.009 (0.017)
High school diploma	-0.068**	-0.068**	-0.110***	-0.111**
	(0.030)	(0.030)	(0.041)	(0.042)
University degree	-0.152***	-0.152***	-0.223***	-0.228***
	(0.037)	(0.037)	(0.045)	(0.045)
Household size	-0.013*	-0.013*	-0.005	-0.005
	(0.008)	(0.008)	(0.016)	(0.016)
Married	-0.120***	-0.120***	-0.044	-0.048
	(0.034)	(0.034)	(0.056)	(0.057)
Self-employed head	0.038	0.039	-0.071	-0.069
	(0.066)	(0.066)	(0.050)	(0.051)
Not-working	0.339***	0.339***	0.172**	0.174**
	(0.051)	(0.051)	(0.075)	(0.077)
Moved from region of birth	0.062** (0.029) 0.109***	0.062** (0.029) 0.108***	0.027 (0.050)	0.028 (0.050)
Living in the Center	(0.036)	(0.036)	0.081	0.073
	0.224***	0.223***	(0.057)	(0.051)
Living in the South Year: 1993	(0.032) 0.220***	(0.032) 0.211***	0.078* (0.043) 0.096**	0.071 (0.043) 0.071
Year: 1995	(0.038)	(0.039)	(0.045)	(0.041)
	0.165***	0.156***	0.075**	0.049
Year: 1998	(0.020)	(0.024)	(0.032)	(0.044)
	0.152***	0.143***	0.065*	0.043
Year: 2000	(0.032) 0.150***	(0.034) 0.143***	(0.036) 0.044	(0.045) (0.045) 0.032
Year: 2002	(0.030) 0.087***	(0.031) 0.085***	(0.045) 0.035	(0.052) (0.052) 0.036
Constant	(0.028)	(0.029)	(0.035)	(0.034)
	1.129***	1.198***	1.335**	1.356**
	(0.168)	(0.181)	(0.554)	(0.565)
Observations	9,563	9,563	1,331	1,331
R-squared	0.114	0.114	0.260	0.263
Marginal propensity to consume	0.905	0.905	0.651	0.655
out of income	(0.202)	(0.202)	(0.131)	(0.136)

Note: See Note to Table 5 for the definition of variables (P<sup>H</sup> is the per-square meter house price). Robust standard errors in parentheses. \*significant at 10%; \*\* significant at 5%; \*\*\* significant at 1%.

	(1)	(2)	(3)	(4)	(5)
	No risky	No risky	Owners of	Owners of	Owners of
	financial	financial	risky fin.	listed stocks	listed stocks
	assets	assets	assets		and real estate
(Net financial wealth)/Y	0.026***	0.026***	0.028***	0.026***	0.021***
	(0.008)	(0.008)	(0.010)	(0.006)	(0.005)
(Net real wealth)/Y	0.019***	0.019***	0.007***	0.007***	0.007***
	(0.003)	(0.003)	(0.001)	(0.001)	(0.001)
$\mathbf{P}_{t+1}^{\mathbf{S}} / \mathbf{P}_{t}^{\mathbf{S}}$		0.052***	-0.056***		
		(0.015)	(0.017)		
Cap. gains on stocks/Y				-0.043	-0.084*
				(0.054)	(0.047)
Cap. gains on real estate/Y		-0.006	-0.006	0.016	0.022*
		(0.007)	(0.008)	(0.012)	(0.012)
Age/10	-0.106***	-0.105***	0.020	0.041	0.049
	(0.025)	(0.026)	(0.021)	(0.037)	(0.039)
Age squared/100	0.004*	0.004*	-0.005**	-0.007*	-0.007*
	(0.002)	(0.002)	(0.002)	(0.004)	(0.004)
High school diploma	-0.050***	-0.049***	-0.050***	-0.040*	-0.037
	(0.008)	(0.008)	(0.016)	(0.022)	(0.024)
University degree	-0.142***	-0.142***	-0.113***	-0.110***	-0.107***
	(0.011)	(0.011)	(0.020)	(0.027)	(0.026)
Household size	-0.017***	-0.017***	-0.006*	-0.004	0.001
	(0.003)	(0.003)	(0.003)	(0.006)	(0.005)
Married	-0.038***	-0.038***	-0.028**	-0.037	-0.017
	(0.009)	(0.009)	(0.013)	(0.024)	(0.019)
Self-employed head	0.017	0.017	-0.019*	-0.003	-0.024
	(0.013)	(0.013)	(0.011)	(0.020)	(0.015)
Not-working	0.129***	0.132***	0.056***	0.062***	0.056***
	(0.018)	(0.018)	(0.011)	(0.017)	(0.017)
Moved from region of birth	0.053***	0.053***	0.037***	0.033**	0.023
	(0.011)	(0.011)	(0.010)	(0.016)	(0.015)
Living in the Center	0.025**	0.024**	0.041***	0.066***	0.061***
	(0.011)	(0.011)	(0.010)	(0.017)	(0.016)
Living in the South	0.067***	0.065***	0.019	0.034	0.017
	(0.007)	(0.007)	(0.014)	(0.027)	(0.015)
Year: 1993	0.063***			0.013	0.010
	(0.011)			(0.021)	(0.022)
Year: 1995	0.066***			0.078***	0.076***
	(0.006)			(0.023)	(0.024)
Year: 1998	0.043***			0.023	0.035*
	(0.009)			(0.020)	(0.020)
Year: 2000	0.040***			0.051**	0.034*
	(0.010)			(0.021)	(0.019)
Year: 2002	0.013*			0.024	0.016
	(0.007)			(0.017)	(0.017)
Constant	1.116***	1.094***	0.770***	0.597***	0.543***
	(0.068)	(0.069)	(0.053)	(0.093)	(0.095)

[Tał	ble 8	continued	1
LIUU		continued	L

Observations	30,879	30,879	6,810	2,528	2,252
R-squared	0.108	0.107	0.173	0.162	0.220
Marginal propensity to consume out of income	0.685	0.685	0.619	0.604	0.592
	(0.088)	(0.085)	(0.052)	(0.0609)	(0.056)

Note: See Note to Table 5 for the definition of variables (P<sup>S</sup> is the stock price index). Robust standard errors in parentheses. \*significant at 10%; \*\* significant at 5%; \*\*\* significant at 1%.

Figure 1

# THE DISTRIBUTIONS OF HOUSE PRICE DYNAMICS ACROSS PROVINCES













## Appendix

#### Data sources

Our main source of data is the Bank of Italy *Survey of Household Income and Wealth* (SHIW), which is a representative sample of the Italian resident population and provides detailed data on household demographic characteristics, consumption, income and balance sheet items. The survey was first run in the mid-1960s but has been available on tape only since 1984. Over time, it has gone through a number of changes in sample size and design, sampling methodology and questionnaire.

For our analysis, we consider the last six surveys of data covering the period 1991-2002, which are broadly homogeneous as to the sampling methodology, the sample size and the contents of the information collected. Over the period considered, the Survey is biannual (with the exception of the 1998 wave which was run three years after the previous one) and covers about 8,000 households per set of interviews. For an exhaustive description of the data, of the sampling methods and issues, see D'Alessio (1993, 1995 and 1997), D'Alessio and Faiella (2000 and 2002) and D'Alessio, Faiella and Neri (2004).

House prices are taken from *Il Consulente Immobilare* (by *Il Sole24Ore*), which collects per-square meter prices of newly-built or recently-renovated houses twice a year. Price data are at provincial level and, for each province, the prices of houses located in the center, semi-center and outskirts are available. For stock prices we use an index of the shares listed on the Italian Stock Exchange, inclusive of price and dividends. Such data were collected directly by the Bank of Italy until 1992; from 1993, we rely on an index of Morgan Stanley (MSCI Italy) which has the same coverage.

## Construction and definitions of the variables

Household <u>net wealth</u> refers to household total assets minus total liabilities as of at the beginning of the year of interview. It is computed subtracting savings and the capital gains/losses on the assets held from end-of-period wealth. Detailed information on end-of-period assets and liabilities is directly available from the data. *Savings* is computed as total after-tax income minus total consumption, as reported in the Survey. The *capital gains/losses* that we consider are those accrued on real estate assets and on listed stocks. Real estate assets include all dwellings, other buildings and land that the household members own at the end of the year of interview. Household <u>net financial</u> and <u>real wealth</u> are computed distinguishing between debt that is related to real estate investment financing and

other liabilities, and assuming that savings are split between real and financial assets as in the national accounts and flow of funds.<sup>16</sup> We then determine beginning-of-period *real estate* and *stocks* assuming that beginning-of-period shares of real assets and financial assets in real estate and stocks, respectively, are the same as at the end of the period. We compute the capital gain/loss accrued on real estate distinguishing, for homeowners, between the house where the household lives and other property. The capital gain/loss on the former is determined as the product of beginning-of-period self-reported value and the change over the coming year in the square-meter price of residential property in the provincial area (central city, semi-central area and the outskirts) where the household lives. The capital gain/loss on the latter is determined as the product of beginning-of-period of beginning-of-period holdings and the change over the coming year in the average square-meter price of residential property in the province where the household lives. We compute the capital gain/loss accrued on stock holdings as the product of beginning-of-period stocks and the change in the province where the household lives. We compute the capital gain/loss accrued on stock holdings as the product of beginning-of-period stocks and the change in the stocks listed on the Italian Stock Exchange. The gains/losses are then netted of the change in consumer prices.

Household <u>expenditure</u> includes spending on non-durable goods and services, including (actual and imputed) rents. It refers to the whole year of interview.

Household earnings includes wages, salaries, pensions and transfers.

<u>Demographic variables</u> refer to the head of the household, unless specified otherwise. *High school diploma* and *University degree* are dummies denoting the highest level of education attained by the head. *Married* and *Self-employed head* are also dummies describing his/her marital status and occupation. *Not-working* is a dummy that takes on value one if the head is unemployed, retired, working at home (house chores) or a student. *Moved from region of birth* is another dummy denoting an individual living in a region which is different from that where he/she was born. *Living in the Center* and *Living in the South* are dummies that take on value one if the head lives in the Center of the country or in the South, which includes the islands, respectively. *Year: 1993* through *Year: 2002* are dummies referring to the year of interview.

We classify as <u>homeowners</u> those households living in a dwelling which is either owned by the household or occupied under redemption agreement. We exclude those who have moved into their house within the year of interview (2.6 percent of all homeowners).

<sup>&</sup>lt;sup>16</sup> Based on the national accounts and on the flow of funds, between 1991 and 1998, 80 percent of financial savings was invested into financial assets; between 2000 and 2002, the share went up to 90 percent.

All monetary variables have been deflated using the Consumer Price Index and are in euros of year 2002.

#### Definition of the sample

From the sample obtained by merging the last six waves of the SHIW, we exclude one household reporting negative non-durable expenditure, 212 reporting negative earnings, 22 households for whom the value of their stock holdings is missing, and other 2,055 households (about 4 percent of the sample) whose head is aged less than 18 or over 80. We also drop 106 households whose capital gains or losses on real estate are over five times their annual earnings and 24 households whose capital losses on stocks are over twice their earnings. Overall, we are left with a sample of 45,147 households.

#### References

- Ando, A., 1988, "Reflections on Some Recent Evidence on Life Cycle Hypothesis of Saving", Studies in Banking and Finance, 5, 7-25.
- Ando, A., L. Guiso and I. Visco (eds.), 1994, Saving and the Accumulation of Wealth: Essays on Italian Household and Government Saving Behavior, Cambridge University Press, Cambridge, U.K.
- Ando, A. and F. Modigliani, 1963, "The 'Life Cycle' Hypothesis of Saving: Aggregate Implications and Tests", *American Economic Review*, 53, 55-84.
- Arena, J.J., 1964, "Capital Gains and the 'Life Cycle' Hypothesis of Saving, American Economic Review, 54, 107-111.
- Attanasio, O., Banks, J. and S. Tanner, 2002, "Asset Holding and Consumption Volatility", *Journal of Political Economy*, 110, 771-792.
- Barca, F., L. Cannari and L. Guiso, 1994, "Bequests and Saving for Retirement. What Imples the Accumulation of Wealth?", in: Ando, A., L. Guiso and I. Visco (eds), Saving and the Accumulation of Wealth: Essays on Italian Household and Government Saving Behavior, Cambridge University Press, Cambridge, U.K., 349-365.
- Bayoumi, T. and H. Edison, 2003, "Is Wealth Increasingly Driving Consumption?", De Nederlandsche Bank, DNB Staff Reports, No. 101.
- Boone, L. and N. Girouard, 2002, "The Stock Market, The Housing Market and Consumer Behavior", OECD Economic Studies, No. 35, 175-200.
- Brandolini, A. and L. Cannari, 1994, "Methodological Appendix: The Bank of Italy's Survey of Household Income and Wealth", in: Ando, A., L. Guiso and I. Visco (eds), Saving and the Accumulation of Wealth: Essays on Italian Household and Government Saving Behavior, Cambridge University Press, Cambridge, U.K., 369-386.
- Carroll, C. and M. Kimball, 1996, "On the Concavity of the Consumption Function", *Econometrica*, 64, 981-992.

- Case, K., J. Quigley and R. Shiller, 2001, "Comparing Wealth Effects: The Stock Market Versus the Housing Market", NBER, Working Paper No. 8606.
- D'Alessio, G., 1993, "I bilanci delle famiglie italiane nell'anno 1991", Banca d'Italia, Supplementi al Bollettino Statistico (nuova serie), No. 44.
- D'Alessio, G., 1995, "I bilanci delle famiglie italiane nell'anno 1993", Banca d'Italia, Supplementi al Bollettino Statistico (nuova serie), No. 9.
- D'Alessio, G., 1997, "I bilanci delle famiglie italiane nell'anno 1995", Banca d'Italia, Supplementi al Bollettino Statistico (nuova serie), No. 14.
- D'Alessio, G. and I. Faiella, 2000, "Italian Households Budgets in 1998", *Supplements to the Statistical Bulletin (new series)*, Bank of Italy, No. 22.
- D'Alessio, G. and I. Faiella, 2002, "Italian Households Budgets in 2000", *Supplements to the Statistical Bulletin (new series)*, Bank of Italy, No. 6.
- D'Alessio, G. and I. Faiella, 2002, "Non-Response Behavior in the Bank of Italy's Survey of Household Income and Wealth", Banca d'Italia, Temi di discussione, No. 462.
- D'Alessio, G., I. Faiella and A. Neri, 2004, "Italian Households Budgets in 2002", Supplements to the Statistical Bulletin (new series), Bank of Italy, No. 12.
- Deaton, A., 1992, Understanding Consumption, Oxford University Press, Oxford.
- Disney, R., A. Henley and D. Jevons, 2003, "House Price Shocks, Negative Equity and Household Consumption in the UK in the 1990s", unpublishedDynan, K., J. Skinner and S. Zeldes, 2000, "Do the Rich Save More?", National Bureau of Economic Research, Working Paper No. 7906.
- Dynan, K. and D. Maki, 2001, "Does Stock Market Wealth Matter for Consumption?", Board of Governors of the Federal Reserve System, FEDS Working Paper No. 2001-21.
- Engelhardt, G., 1996, "House Prices and Home Owner Saving Behavior", *Regional Science* & Urban Economics, 26, 313-336.
- Friedman, M., 1957, *A Theory of the Consumption Function*, Princeton University Press, Princeton.

- Goodman, J. and J.B. Ittner, 1992, "The Accuracy of Homeowners' Estimates of House Value", *Journal of Housing Economics*, 2, 339-57.
- Guiso, L., M. Haliassos and T. Jappelli (eds.), 2001, *Household Portfolios*, MIT Press, Cambridge, Mass.
- Kain, J. and J. Quigley, 1972, "Note on Owners' Estimates of Housing Values", *Journal of American Statistical Association*, 67, 803-6.
- King, M. and J. Leape, 1998, "Wealth and Portfolio Composition: Theory and Evidence", *Journal of Public Economics*, 69, 155-193.
- Hoynes, H. and D. McFadden, 1997, "The Impact of Demographics on Housing and Nonhousing Wealth in the United States", in: Hurd, M. and N. Yashiro (eds.), *The Economic Effects of Aging in the United States and Japan*, University of Chicago Press, Chicago,153-194.
- Levin, L., 1998, "Are Assets Fungible? Testing the Behavioral Theory of Life-Cycle Savings", *Journal of Economic Behavior & Organization*, 36, 59-83.
- Maki, D. and M. Palumbo, 2001, "Disentangling the Wealth Effect: A Cohort Analysis of Household Savings in the 1990s", Board of Governors of the Federal Reserve, FEDS Working Paper No. 2001-23.
- Mankiw, N. and S. Zeldes, 1991, "The Consumption of Stockholders and Non-Stockholders", *Journal of Financial Economics*, 17, 211-19.
- Modigliani, F., 1975, "The Life Cycle Hypothesis of Saving Twenty Years Later", in: Parkin, M. and A.R. Nobay, *Contemporary Issues in Economics*, Manchester University Press, Manchester, 2-36.
- Modigliani, F. and A. Ando, 1960, "The Permanent Income and the Life Cycle Hypothesis of Saving Behavior: Comparisons and Test", in: Friend, I. and R. Jones (eds.), *Proceedings of the Conference on Consumption and Saving*, vol. II, University of Pennsylvania, Philadelphia, 49-174.
- Modigliani, F. and R. Brumberg, 1954, "Utility Analysis and the consumption Function: An Interpretation of Cross-Section Data", in: Kurihara, K.K. (ed.), *Post-Keynesian Economics*, Rutgers University Press, New Brunswick, N.J., 388-436.

- Modigliani F. and R. Brumberg, 1980 [1954], "Utility Analysis and Aggregate Consumption Functions: An Attempt at Integration", in: Abel, A. (ed.), *The Collected Papers of Franco Modigliani*, The MIT Press, Cambridge, Mass., 128-197.
- Paiella, M., 2004, "Revisiting the Implications of Heterogeneity in Financial Market Participation for the C-CAPM", *European Finance Review*, 8, 1-36.
- Poterba, J., 2000, "Stock Market Wealth and Consumption", *Journal of Economic Perspectives*, 14, 99–118.
- Poterba, J. and A. Samwick, 1995, "Stock Ownership Patterns, Stock Market Fluctuations and Consumption", *Brookings Papers on Economic Activity*, 2, 295-371.
- Sheiner, L., 1995, "Housing Prices and the Saving Rate of Renters", Journal of Urban Economics, 38, 94-125.
- Skinner, J., 1996, "Is Housing Wealth a Sideshow?", in: Wise, D.A. (ed.), Advances in the Economics of Aging, A National Bureau of Economic Research Project Report, The University of Chicago Press, Chicago, 242-271.
- Starr-McCluer, M., 1998, "Stock Market Wealth and Consumer Spending", Board of Governors of the Federal Reserve System, FEDS Working Paper No. 1998-20.
- Thaler, R., 1990, "Anomalies: Saving, Fungibility and Mental Accounts", *Journal of Economic Perspectives*, 4, 93-206.
- Tobin, J., 1967, "Life Cycle Saving and Balanced Growth", in: Fellner, W. (ed.), *Ten Economic Studies in the Tradition of Irving Fisher*, Wiley, New York, 231-256.
- Tobin, J. and D.C. Dolde, 1971, "Wealth, Liquidity and Consumption", in: Federal Reserve Bank of Boston, *Consumer Spending and Monetary Policy: The Linkages*, Conference Series No. 5, 99-146.
- Vissing-Jørgensen, A., 2002, "Limited Asset Market Participation and the Elasticity of Intertemporal Substitution", *Journal of Political Economy*, 110, 825-853.
- White, H., 1980, "A Heteroskedasticity-Consistent Covariance Matrix Estimator and a Direct Test for Heteroskedasticity", *Econometrica*, 48, 817-38.

Zeldes, S., 1989, "Optimal Consumption with Stochastic Income: Deviations from Certainty Equivalence", *Quarterly Journal of Economics*, 104, 275-298.

- N. 530 *Household wealth distribution in Italy in the 1990s*, by A. BRANDOLINI, L. CANNARI, G. D'ALESSIO and I. FAIELLA (December 2004).
- N. 531 Cyclical asymmetry in fiscal policy, debt accumulation and the Treaty of Maastricht, by F. BALASSONE and M. FRANCESE (December 2004).
- N. 532 L'introduzione dell'euro e la divergenza tra inflazione rilevata e percepita, by P. DEL GIOVANE and R. SABBATINI (December 2004).
- N. 533 A micro simulation model of demographic development and households' economic behavior in Italy, by A. ANDO and S. NICOLETTI ALTIMARI (December 2004).
- N. 534 Aggregation bias in macro models: does it matter for the euro area?, by L. MONTEFORTE (December 2004).
- N. 535 Entry decisions and adverse selection: an empirical analysis of local credit markets, by G. GOBBI and F. LOTTI (December 2004).
- N. 536 An empirical investigation of the relationship between inequality and growth, by P. PAGANO (December 2004).
- N. 537 Monetary policy impulses, local output and the transmission mechanism, by M. CARUSO (December 2004).
- N. 538 An empirical micro matching model with an application to Italy and Spain, by F. PERACCHI and E.VIVIANO (December 2004).
- N. 539 La crescita dell'economia italiana negli anni novanta tra ritardo tecnologico e rallentamento della produttività, by A. BASSANETTI, M. IOMMI, C. JONA-LASINIO and F. ZOLLINO (December 2004).
- N. 540 Cyclical sensitivity of fiscal policies based on real-time data, by L. FORNI and S. MOMIGLIANO (December 2004).
- N. 541 L'introduzione dell'euro e le politiche di prezzo: analisi di un campione di dati individuali, by E. GAIOTTI and F. LIPPI (February 2005).
- N. 542 How do banks set interest rates?, by L. GAMBACORTA (February 2005).
- N. 543 Maxmin portfolio choice, by M. TABOGA (February 2005).
- N. 544 Forecasting output growth and inflation in the euro area: are financial spreads useful?, by A. NOBILI (February 2005).
- N. 545 Can option smiles forecast changes in interest rates? An application to the US, the UK and the Euro Area, by M. PERICOLI (February 2005).
- N. 546 The role of risk aversion in predicting individual behavior, by L. Guiso and M. PAIELLA (February 2005).
- N. 547 Prices, product differentiation and quality measurement: a comparison between hedonic and matched model methods, by G. M. TOMAT (February 2005).
- N. 548 The Basel Committee approach to risk weights and external ratings: what do we learn from bond spreads?, by A. RESTI and A. SIRONI (February 2005).
- N. 549 Firm size distribution: do financial constraints explain it all? Evidence from survey data, by P. ANGELINI and A. GENERALE (June 2005).
- N. 550 Proprietà, controllo e trasferimenti nelle imprese italiane. Cosa è cambiato nel decennio 1993-2003?, by S. GIACOMELLI and S. TRENTO (June 2005).
- N. 551 Quota dei Profitti e redditività del capitale in Italia: un tentativo di interpretazione, by R. TORRINI (June 2005).
- N. 552 *Hiring incentives and labour force participation in Italy*, by P. CIPOLLONE, C. DI MARIA and A. GUELFI (June 2005).
- N. 553 Trade credit as collateral, by M. OMICCIOLI (June 2005).
- N. 554 Where do human capital externalities end up?, by A. DALMAZZO and G. DE BLASIO (June 2005).

<sup>(\*)</sup> Requests for copies should be sent to:

Banca d'Italia – Servizio Studi – Divisione Biblioteca e pubblicazioni – Via Nazionale, 91 – 00184 Rome (fax 0039 06 47922059). They are available on the Internet www.bancaditalia.it.

- L. GUISO and G. PARIGI, *Investment and demand uncertainty*, Quarterly Journal of Economics, Vol. 114 (1), pp. 185-228, **TD No. 289** (November 1996).
- A. F. POZZOLO, *Gli effetti della liberalizzazione valutaria sulle transazioni finanziarie dell'Italia con l'estero*, Rivista di Politica Economica, Vol. 89 (3), pp. 45-76, **TD No. 296 (February 1997)**.
- A. CUKIERMAN and F. LIPPI, Central bank independence, centralization of wage bargaining, inflation and unemployment: theory and evidence, European Economic Review, Vol. 43 (7), pp. 1395-1434, TD No. 332 (April 1998).
- P. CASELLI and R. RINALDI, La politica fiscale nei paesi dell'Unione europea negli anni novanta, Studi e note di economia, (1), pp. 71-109, TD No. 334 (July 1998).
- A. BRANDOLINI, The distribution of personal income in post-war Italy: Source description, data quality, and the time pattern of income inequality, Giornale degli economisti e Annali di economia, Vol. 58 (2), pp. 183-239, TD No. 350 (April 1999).
- L. GUISO, A. K. KASHYAP, F. PANETTA and D. TERLIZZESE, Will a common European monetary policy have asymmetric effects?, Economic Perspectives, Federal Reserve Bank of Chicago, Vol. 23 (4), pp. 56-75, TD No. 384 (October 2000).

- P. ANGELINI, Are banks risk-averse? Timing of the operations in the interbank market, Journal of Money, Credit and Banking, Vol. 32 (1), pp. 54-73, **TD No. 266 (April 1996).**
- F. DRUDI and R: GIORDANO, *Default Risk and optimal debt management*, Journal of Banking and Finance, Vol. 24 (6), pp. 861-892, **TD No. 278 (September 1996)**.
- F. DRUDI and R. GIORDANO, *Wage indexation, employment and inflation,* Scandinavian Journal of Economics, Vol. 102 (4), pp. 645-668, **TD No. 292 (December 1996)**.
- F. DRUDI and A. PRATI, *Signaling fiscal regime sustainability*, European Economic Review, Vol. 44 (10), pp. 1897-1930, **TD No. 335 (September 1998)**.
- F. FORNARI and R. VIOLI, The probability density function of interest rates implied in the price of options, in: R. Violi, (ed.), Mercati dei derivati, controllo monetario e stabilità finanziaria, Il Mulino, Bologna, TD No. 339 (October 1998).
- D. J. MARCHETTI and G. PARIGI, Energy consumption, survey data and the prediction of industrial production in Italy, Journal of Forecasting, Vol. 19 (5), pp. 419-440, TD No. 342 (December 1998).
- A. BAFFIGI, M. PAGNINI and F. QUINTILIANI, Localismo bancario e distretti industriali: assetto dei mercati del credito e finanziamento degli investimenti, in: L.F. Signorini (ed.), Lo sviluppo locale: un'indagine della Banca d'Italia sui distretti industriali, Donzelli, TD No. 347 (March 1999).
- A. SCALIA and V. VACCA, Does market transparency matter? A case study, in: Market Liquidity: Research Findings and Selected Policy Implications, Basel, Bank for International Settlements, TD No. 359 (October 1999).
- F. SCHIVARDI, *Rigidità nel mercato del lavoro, disoccupazione e crescita*, Giornale degli economisti e Annali di economia, Vol. 59 (1), pp. 117-143, **TD No. 364 (December 1999)**.
- G. BODO, R. GOLINELLI and G. PARIGI, *Forecasting industrial production in the euro area*, Empirical Economics, Vol. 25 (4), pp. 541-561, **TD No. 370 (March 2000)**.
- F. ALTISSIMO, D. J. MARCHETTI and G. P. ONETO, *The Italian business cycle: Coincident and leading indicators and some stylized facts*, Giornale degli economisti e Annali di economia, Vol. 60 (2), pp. 147-220, **TD No. 377 (October 2000)**.

- C. MICHELACCI and P. ZAFFARONI, (*Fractional*) *Beta convergence*, Journal of Monetary Economics, Vol. 45, pp. 129-153, **TD No. 383 (October 2000)**.
- R. DE BONIS and A. FERRANDO, The Italian banking structure in the nineties: testing the multimarket contact hypothesis, Economic Notes, Vol. 29 (2), pp. 215-241, TD No. 387 (October 2000).

- M. CARUSO, Stock prices and money velocity: A multi-country analysis, Empirical Economics, Vol. 26 (4), pp. 651-72, TD No. 264 (February 1996).
- P. CIPOLLONE and D. J. MARCHETTI, *Bottlenecks and limits to growth: A multisectoral analysis of Italian industry*, Journal of Policy Modeling, Vol. 23 (6), pp. 601-620, **TD No. 314 (August 1997)**.
- P. CASELLI, Fiscal consolidations under fixed exchange rates, European Economic Review, Vol. 45 (3), pp. 425-450, TD No. 336 (October 1998).
- F. ALTISSIMO and G. L. VIOLANTE, Nonlinear VAR: Some theory and an application to US GNP and unemployment, Journal of Applied Econometrics, Vol. 16 (4), pp. 461-486, TD No. 338 (October 1998).
- F. NUCCI and A. F. POZZOLO, *Investment and the exchange rate*, European Economic Review, Vol. 45 (2), pp. 259-283, **TD No. 344 (December 1998)**.
- L. GAMBACORTA, On the institutional design of the European monetary union: Conservatism, stability pact and economic shocks, Economic Notes, Vol. 30 (1), pp. 109-143, **TD No. 356 (June 1999)**.
- P. FINALDI RUSSO and P. ROSSI, Credit costraints in italian industrial districts, Applied Economics, Vol. 33 (11), pp. 1469-1477, TD No. 360 (December 1999).
- A. CUKIERMAN and F. LIPPI, *Labor markets and monetary union: A strategic analysis*, Economic Journal, Vol. 111 (473), pp. 541-565, **TD No. 365 (February 2000)**.
- G. PARIGI and S. SIVIERO, An investment-function-based measure of capacity utilisation, potential output and utilised capacity in the Bank of Italy's quarterly model, Economic Modelling, Vol. 18 (4), pp. 525-550, TD No. 367 (February 2000).
- F. BALASSONE and D. MONACELLI, *Emu fiscal rules: Is there a gap?*, in: M. Bordignon and D. Da Empoli (eds.), Politica fiscale, flessibilità dei mercati e crescita, Milano, Franco Angeli, **TD No.** 375 (July 2000).
- A. B. ATKINSON and A. BRANDOLINI, Promise and pitfalls in the use of "secondary" data-sets: Income inequality in OECD countries, Journal of Economic Literature, Vol. 39 (3), pp. 771-799, TD No. 379 (October 2000).
- D. FOCARELLI and A. F. POZZOLO, The determinants of cross-border bank shareholdings: An analysis with bank-level data from OECD countries, Journal of Banking and Finance, Vol. 25 (12), pp. 2305-2337, TD No. 381 (October 2000).
- M. SBRACIA and A. ZAGHINI, *Expectations and information in second generation currency crises models*, Economic Modelling, Vol. 18 (2), pp. 203-222, **TD No. 391 (December 2000)**.
- F. FORNARI and A. MELE, Recovering the probability density function of asset prices using GARCH as diffusion approximations, Journal of Empirical Finance, Vol. 8 (1), pp. 83-110, TD No. 396 (February 2001).
- P. CIPOLLONE, La convergenza dei salari manifatturieri in Europa, Politica economica, Vol. 17 (1), pp. 97-125, **TD No. 398 (February 2001)**.
- E. BONACCORSI DI PATTI and G. GOBBI, The changing structure of local credit markets: Are small businesses special?, Journal of Banking and Finance, Vol. 25 (12), pp. 2209-2237, TD No. 404 (June 2001).
- G. MESSINA, Decentramento fiscale e perequazione regionale. Efficienza e redistribuzione nel nuovo sistema di finanziamento delle regioni a statuto ordinario, Studi economici, Vol. 56 (73), pp. 131-148, TD No. 416 (August 2001).

- R. CESARI and F. PANETTA, Style, fees and performance of Italian equity funds, Journal of Banking and Finance, Vol. 26 (1), TD No. 325 (January 1998).
- L. GAMBACORTA, Asymmetric bank lending channels and ECB monetary policy, Economic Modelling, Vol. 20 (1), pp. 25-46, **TD No. 340 (October 1998)**.
- C. GIANNINI, "Enemy of none but a common friend of all"? An international perspective on the lenderof-last-resort function, Essay in International Finance, Vol. 214, Princeton, N. J., Princeton University Press, TD No. 341 (December 1998).
- A. ZAGHINI, Fiscal adjustments and economic performing: A comparative study, Applied Economics, Vol. 33 (5), pp. 613-624, TD No. 355 (June 1999).
- F. ALTISSIMO, S. SIVIERO and D. TERLIZZESE, *How deep are the deep parameters?*, Annales d'Economie et de Statistique,.(67/68), pp. 207-226, **TD No. 354 (June 1999)**.
- F. FORNARI, C. MONTICELLI, M. PERICOLI and M. TIVEGNA, *The impact of news on the exchange rate of the lira and long-term interest rates*, Economic Modelling, Vol. 19 (4), pp. 611-639, **TD No. 358** (October 1999).
- D. FOCARELLI, F. PANETTA and C. SALLEO, *Why do banks merge?*, Journal of Money, Credit and Banking, Vol. 34 (4), pp. 1047-1066, **TD No. 361 (December 1999)**.
- D. J. MARCHETTI, Markup and the business cycle: Evidence from Italian manufacturing branches, Open Economies Review, Vol. 13 (1), pp. 87-103, **TD No. 362 (December 1999)**.
- F. BUSETTI, *Testing for stochastic trends in series with structural breaks*, Journal of Forecasting, Vol. 21 (2), pp. 81-105, TD No. 385 (October 2000).
- F. LIPPI, *Revisiting the Case for a Populist Central Banker*, European Economic Review, Vol. 46 (3), pp. 601-612, **TD No. 386 (October 2000)**.
- F. PANETTA, The stability of the relation between the stock market and macroeconomic forces, Economic Notes, Vol. 31 (3), TD No. 393 (February 2001).
- G. GRANDE and L. VENTURA, Labor income and risky assets under market incompleteness: Evidence from Italian data, Journal of Banking and Finance, Vol. 26 (2-3), pp. 597-620, TD No. 399 (March 2001).
- A. BRANDOLINI, P. CIPOLLONE and P. SESTITO, *Earnings dispersion, low pay and household poverty in Italy, 1977-1998*, in D. Cohen, T. Piketty and G. Saint-Paul (eds.), The Economics of Rising Inequalities, pp. 225-264, Oxford, Oxford University Press, **TD No. 427** (November 2001).
- L. CANNARI and G. D'ALESSIO, La distribuzione del reddito e della ricchezza nelle regioni italiane, Rivista Economica del Mezzogiorno (Trimestrale della SVIMEZ), Vol. XVI (4), pp. 809-847, Il Mulino, TD No. 482 (June 2003).

- F. SCHIVARDI, *Reallocation and learning over the business cycle,* European Economic Review, , Vol. 47 (1), pp. 95-111, **TD No. 345 (December 1998)**.
- P. CASELLI, P. PAGANO and F. SCHIVARDI, Uncertainty and slowdown of capital accumulation in Europe, Applied Economics, Vol. 35 (1), pp. 79-89, **TD No. 372 (March 2000).**
- P. ANGELINI and N. CETORELLI, *The effect of regulatory reform on competition in the banking industry*, Federal Reserve Bank of Chicago, Journal of Money, Credit and Banking, Vol. 35, pp. 663-684, **TD No. 380 (October 2000)**.

- P. PAGANO and G. FERRAGUTO, Endogenous growth with intertemporally dependent preferences, Contribution to Macroeconomics, Vol. 3 (1), pp. 1-38, **TD No. 382 (October 2000).**
- P. PAGANO and F. SCHIVARDI, *Firm size distribution and growth*, Scandinavian Journal of Economics, Vol. 105 (2), pp. 255-274, **TD No. 394 (February 2001)**.
- M. PERICOLI and M. SBRACIA, A Primer on Financial Contagion, Journal of Economic Surveys, Vol. 17 (4), pp. 571-608, TD No. 407 (June 2001).
- M. SBRACIA and A. ZAGHINI, *The role of the banking system in the international transmission of shocks*, World Economy, Vol. 26 (5), pp. 727-754, **TD No. 409 (June 2001)**.
- E. GAIOTTI and A. GENERALE, Does monetary policy have asymmetric effects? A look at the investment decisions of Italian firms, Giornale degli Economisti e Annali di Economia, Vol. 61 (1), pp. 29-59, TD No. 429 (December 2001).
- L. GAMBACORTA, The Italian banking system and monetary policy transmission: evidence from bank level data, in: I. Angeloni, A. Kashyap and B. Mojon (eds.), Monetary Policy Transmission in the Euro Area, Cambridge, Cambridge University Press, TD No. 430 (December 2001).
- M. EHRMANN, L. GAMBACORTA, J. MARTÍNEZ PAGÉS, P. SEVESTRE and A. WORMS, *Financial systems and the role of banks in monetary policy transmission in the euro area*, in: I. Angeloni, A. Kashyap and B. Mojon (eds.), Monetary Policy Transmission in the Euro Area, Cambridge, Cambridge University Press, **TD No. 432 (December 2001)**.
- F. SPADAFORA, Financial crises, moral hazard and the speciality of the international market: further evidence from the pricing of syndicated bank loans to emerging markets, Emerging Markets Review, Vol. 4 (2), pp. 167-198, TD No. 438 (March 2002).
- D. FOCARELLI and F. PANETTA, Are mergers beneficial to consumers? Evidence from the market for bank deposits, American Economic Review, Vol. 93 (4), pp. 1152-1172, **TD No. 448 (July 2002)**.
- E.VIVIANO, Un'analisi critica delle definizioni di disoccupazione e partecipazione in Italia, Politica Economica, Vol. 19 (1), pp. 161-190, **TD No. 450** (July 2002).
- M. PAGNINI, Misura e Determinanti dell'Agglomerazione Spaziale nei Comparti Industriali in Italia, Rivista di Politica Economica, Vol. 3 (4), pp. 149-196, **TD No. 452 (October 2002)**.
- F. BUSETTI and A. M. ROBERT TAYLOR, Testing against stochastic trend and seasonality in the presence of unattended breaks and unit roots, Journal of Econometrics, Vol. 117 (1), pp. 21-53, TD No. 470 (February 2003).

- F. LIPPI, Strategic monetary policy with non-atomistic wage-setters, Review of Economic Studies, Vol. 70 (4), pp. 909-919, TD No. 374 (June 2000).
- P. CHIADES and L. GAMBACORTA, *The Bernanke and Blinder model in an open economy: The Italian case*, German Economic Review, Vol. 5 (1), pp. 1-34, **TD No. 388 (December 2000)**.
- M. BUGAMELLI and P. PAGANO, *Barriers to Investment in ICT*, Applied Economics, Vol. 36 (20), pp. 2275-2286, **TD No. 420 (October 2001)**.
- A. BAFFIGI, R. GOLINELLI and G. PARIGI, *Bridge models to forecast the euro area GDP*, International Journal of Forecasting, Vol. 20 (3), pp. 447-460, **TD No. 456 (December 2002)**.
- D. AMEL, C. BARNES, F. PANETTA and C. SALLEO, Consolidation and Efficiency in the Financial Sector: A Review of the International Evidence, Journal of Banking and Finance, Vol. 28 (10), pp. 2493-2519, TD No. 464 (December 2002).
- M. PAIELLA, *Heterogeneity in financial market participation: appraising its implications for the C-CAPM*, Review of Finance, Vol. 8, pp. 1-36, **TD No. 473 (June 2003)**.
- E. BARUCCI, C. IMPENNA and R. RENÒ, *Monetary integration, markets and regulation*, Research in Banking and Finance, (4), pp. 319-360, **TD No. 475** (June 2003).

- E. BONACCORSI DI PATTI and G. DELL'ARICCIA, *Bank competition and firm creation*, Journal of Money Credit and Banking, Vol. 36 (2), pp. 225-251, **TD No. 481 (June 2003)**.
- R. GOLINELLI and G. PARIGI, Consumer sentiment and economic activity: a cross country comparison, Journal of Business Cycle Measurement and Analysis, Vol. 1 (2), pp. 147-172, TD No. 484 (September 2003).
- L. GAMBACORTA and P. E. MISTRULLI, *Does bank capital affect lending behavior?*, Journal of Financial Intermediation, Vol. 13 (4), pp. 436-457, **TD No. 486 (September 2003)**.
- F. SPADAFORA, Il pilastro privato del sistema previdenziale: il caso del Regno Unito, Rivista Economia Pubblica, (5), pp. 75-114, TD No. 503 (June 2004).
- G. GOBBI e F. LOTTI, Entry decisions and adverse selection: an empirical analysis of local credit markets, Journal of Financial services Research, Vol. 26 (3), pp. 225-244, TD No. 535 (December 2004).
- F. CINGANO e F. SCHIVARDI, *Identifying the sources of local productivity growth*, Journal of the European Economic Association, Vol. 2 (4), pp. 720-742, **TD No. 474 (June 2003)**.
- C. BENTIVOGLI e F. QUINTILIANI, Tecnologia e dinamica dei vantaggi comparati: un confronto fra quattro regioni italiane, in C. Conigliani (a cura di), Tra sviluppo e stagnazione: l'economia dell'Emilia-Romagna, Bologna, Il Mulino, **TD No. 522** (October 2004).

- A. DI CESARE, Estimating Expectations of Shocks Using Option Prices, The ICFAI Journal of Derivatives Markets, Vol. II (1), pp. 42-53, TD No. 506 (July 2004).
- M. OMICCIOLI, Il credito commerciale: problemi e teorie, in L. Cannari, S. Chiri e M. Omiccioli (a cura di), Imprese o intermediari? Aspetti finanziari e commerciali del credito tra imprese in Italia, Bologna, Il Mulino, TD No. 494 (June 2004).
- L. CANNARI, S. CHIRI e M. OMICCIOLI, Condizioni del credito commerciale e differenzizione della clientela, in L. Cannari, S. Chiri e M. Omiccioli (a cura di), Imprese o intermediari? Aspetti finanziari e commerciali del credito tra imprese in Italia, Bologna, Il Mulino, TD No. 495 (June 2004).
- P. FINALDI RUSSO e L. LEVA, Il debito commerciale in Italia: quanto contano le motivazioni finanziarie?, in L. Cannari, S. Chiri e M. Omiccioli (a cura di), Imprese o intermediari? Aspetti finanziari e commerciali del credito tra imprese in Italia, Bologna, Il Mulino, TD No. 496 (June 2004).
- A. CARMIGNANI, Funzionamento della giustizia civile e struttura finanziaria delle imprese: il ruolo del credito commerciale, in L. Cannari, S. Chiri e M. Omiccioli (a cura di), Imprese o intermediari? Aspetti finanziari e commerciali del credito tra imprese in Italia, Bologna, Il Mulino, TD No. 497 (June 2004).
- G. DE BLASIO, Does trade credit substitute for bank credit?, in L. Cannari, S. Chiri e M. Omiccioli (a cura di), Imprese o intermediari? Aspetti finanziari e commerciali del credito tra imprese in Italia, Bologna, Il Mulino, TD No. 498 (June 2004).
- M. BENVENUTI e M. GALLO, Perché le imprese ricorrono al factoring? Il caso dell'Italia, in L. Cannari, S. Chiri e M. Omiccioli (a cura di), Imprese o intermediari? Aspetti finanziari e commerciali del credito tra imprese in Italia, Bologna, Il Mulino, TD No. 518 (October 2004).